Author's response to reviews

Title: Is there an impact of public smoking bans on self-reported smoking status and exposure to secondhand smoke? Do smoking bans decrease exposure?

Authors:

    Alisa B Naiman (a.naiman@utoronto.ca)
    Richard H Glazier (rick.glazier@ices.on.ca)
    Rahim Moineddin (rahim.moineddin@utoronto.ca)

Version: 8 Date: 28 February 2011

Author's response to reviews:

Please find attached a further revision
reference 18 has been deleted